Author’s response to reviews

Title: Antituberculosis activity, phytochemical identification of Costus speciosus (J. Koenig) Sm., Cymbopogon citratus (DC. ex Nees) Stapf., and Tabernaemontana coronaria (L.) Willd. and their effects on the growth kinetics and cellular integrity of Mycobacterium tuberculosis H37Rv

Authors:

Suriyati Mohamad (suri@usm.my)
Nur Najihah Ismail (norjihah@yahoo.com)
Thaigarajan Parumasivam (thaigarp@usm.my)
Pazilah Ibrahim (pazilah@usm.my)
Hasnah Osman (ohasnah@usm.my)
Habibah Wahab (habibahw@usm.my)

Version: 2 Date: 23 Dec 2017

Author’s response to reviews:

Please make the following revisions to your manuscript:

1. Please only include one copy of the finalized tables. (Done)

2. Please move all figure legends to the end of the manuscript in a section labeled, "Figure legends". (Done)

3. Please delete all of your supplementary files. They are no longer necessary at this stage of production. (Done)

4. Upon uploading your revisions, please remove any tracked changes or highlighting and include only a single clean copy of the manuscript. (Done)